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NuMI Beam Status

• Target Hall air chiller compressor work continues:
– New compressor installed last week 
– Add refrigerant (today). Restart chiller (tomorrow)
– Run beam evenings and nights
– Plan to resume normal beam on Tuesday pm

• Target scan last Thursday. Looks normal.



NuMI Protons Delivered (last week)

Integrated protons on target 3.6E18



Protons Delivered/Day

Daily Protons Delivered to NuMI Target Starting 1 May 2005 (measured by CALLIBRATION CORRECTED values for E:TORTGT)
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Detectors Status 

• Far detector (99.32% up-time):
– HV down (MF4) for few hours on Saturday night. 

Controller board has been replaced. Fixed now.
– 10 min network downtime planned for this afternoon.

• Near detector (99.76% up-time):
– Electronics maintenance: 5 minder cards and 1 master 

card replaced. 


